Author's response to reviews

Title: Emergence of Serogroup C Meningococcal Disease Associated with a High Mortality Rate in Hefei, China

Authors:

Xi-Hai Xu (2xhai@163.com)
Ying Ye (yeying2009@163.com)
Li-Fen Hu (lifenhu@gmail.com)
Qin-Qin Jiang (jiangqq2010@163.com)
Yu-Hui Jin (yhujin@sina.com.cn)
Jia-Bin Li (jiabinli948@163.com)

Version: 6 Date: 13 August 2012

Author's response to reviews: see over
Dear Dr. Philippa Harris,

Thank you very much for your patient and advice on our paper. We have asked helps from a professional language editing service (Edanz) to copyedit our manuscript to improve the style of written English. The amendments are highlighted in the revised manuscript. We also have correctly formatted our revised manuscript according to the journal style. We learn a lot from this experience and hope the revision is acceptable. We are looking forward to hearing from you soon.

With best wishes,

Yours sincerely,

Jia-Bin Li

Department of Infectious Diseases,

the First Affiliated Hospital of Anhui Medical University,

Hefei, Anhui 230022, the People’s Republic of China.